Author's response to reviews

Title: Using a practical molecular capsular serotype prediction strategy to investigate Streptococcus pneumoniae serotype distribution and antimicrobial resistance in Chinese local hospitalized children

Authors:

Ping Jin (yanyanab2004@aliyun.com)
Lijuan Wu (6345650@qq.com)
Shahin Oftadeh (Shahin.Oftadeh@health.nsw.gov.au)
Timothy Kudinha (tkudinha@yahoo.com)
Fanrong Kong (Fanrong.Kong@health.nsw.gov.au)
Qiyi Zeng (qiyizengshenzhen@163.com)

Version: 10 Date: 15 February 2016

Author's response to reviews: see over
Reply to the editor’ comments

Date: 14 February 2016

Dr Catherine Olino
BMC Pediatrics
Journal Editorial Office BioMed Central

Dear Dr Olino,

RE: MS: 1670154452176925 - Using a practical molecular capsular serotype prediction strategy to investigate Streptococcus pneumoniae serotype distribution and antimicrobial resistance in Chinese local hospitalized children

Thank you for the opportunity to revise our manuscript. We are pleased to address the editor’s comments point-by-point as below.

Question. We notice you have deposited your additional files in a repository. We kindly ask that you instead upload these as separate additional files to your manuscript.

Answer: Thanks for your suggestion. We have submitted it as additional files to my manuscript.

Thank you for taking the time to consider this manuscript for publication.

Yours sincerely,

Professor Qiyi Zeng